Author's response to reviews

Title: Irradiation-induced telomerase activity and gastric cancer risk: a case-control analysis in a Chinese Han population

Authors:

Xianli He (wanghe@fmmu.edu.cn)
Qing Qiao (qging@gmail.com)
Naijian Ge (gelarge@163.com)
Jing Nan (nanjing0507@163.com)
Shuqun Shen (shenshuqunssq@sohu.com)
Zizhong Zhang (xskxue@fmmu.edu.cn)
Yefa Yang (TISC@EHBH.cn)
Guoqiang Bao (guoqiang@fmmu.edu.cn)

Version: 6 Date: 14 June 2010

Author's response to reviews:

Dear editor,

The major revisions have been made according to your request.

In the manuscript, chinese people's names are expressed in westernized style(first name=given name; last name=family name). Chinese people generally have no middle name. The style match with the current ones used in the published articles.

Sincerely,

Guoqiang Bao